
-----------------------------------------------------------------------------
                Mauna Loa Solar Observatory Observer’s Log 
-----------------------------------------------------------------------------
        Mon Jul 11 17:03:53 GMT 2005
Year: 05  Doy: 192
Observer: yasukawa
WEATHER COMMENT: Mon Jul 11 17:05:17 GMT 2005
In altostratus overcast and low clouds, light southeast breeze, temp 46 F.
R.H. 85%
Keeping domes closed until weather improves.
___end___
ECHO COMMENT: Mon Jul 11 17:07:26 GMT 2005
Time OK.
___end___
PSPT COMMENT: Mon Jul 11 18:16:25 GMT 2005
Not sure if pspt was run yesterday, but tape was popped out so
I moved it to kenobi to transfer any data that may be on it.
There could be old data -- please check.
___end___
A NEW PSPT EXABYTE IS LOADED ON KENOBI  Mon Jul 11 18:18:11 GMT 2005
GONG COMMENT: Mon Jul 11 19:03:34 GMT 2005
Going down to GONG to take care of the HD swap and to do weekly PM.
___end___
Mon Jul 11 20:12:44 GMT 2005    MKIV    Start Patrol
Mon Jul 11 20:13:07 GMT 2005    CHIP     Start Patrol
Mon Jul 11 20:13:23 GMT 2005    PICS     Start Patrol
WEATHER COMMENT: Mon Jul 11 20:13:09 GMT 2005
Clouds breaking up.  Opened domes and starting observations.
___end___
Mon Jul 11 20:19:03 GMT 2005    MKIV    End Patrol
PSPT COMMENT: Mon Jul 11 20:28:02 GMT 2005
Observing.
___end___
Mon Jul 11 20:38:41 GMT 2005    MKIV    End Cal
WEATHER COMMENT: Mon Jul 11 20:38:45 GMT 2005
In cloud, idling mk4.
___end___
NICE MK4      IMAGE: 2012
NICE CHIP     IMAGE: 2013
NICE PICSLIMB IMAGE: 2014
NICE PICSDISC IMAGE: 2013
Mon Jul 11 20:40:33 GMT 2005    MKIV    Start Patrol
WEATHER COMMENT: Mon Jul 11 20:40:37 GMT 2005
In hole, resuming mk4 obs.
___end___
**CHIP PROBLEM**: Mon Jul 11 21:40:18 GMT 2005
CHIP stalled -- segmentation fault.  Killing patrol and restarting Run.
___end___



Mon Jul 11 21:43:06 GMT 2005    CHIP     Start Patrol
Mon Jul 11 21:54:31 GMT 2005    CHIP     End Patrol
Mon Jul 11 21:54:43 GMT 2005    PICS     End Patrol
WEATHER COMMENT: Mon Jul 11 21:55:09 GMT 2005
In thick clouds, holes closing up fast.  Closing domes as a precaution
against rain.
___end___
Mon Jul 11 21:57:09 GMT 2005    MKIV    End Patrol
**GONG PROBLEM**: Mon Jul 11 23:44:45 GMT 2005
Going down to GONG for the 5th time today.  This time to reboot the
workstation.  They keep coming up with these tasks one at a time rather
than consolidating them. 
___end___
**GONG PROBLEM**: Tue Jul 12 00:17:38 GMT 2005
Back from GONG.  Reboot did not get rid of "NO WCS" message on ximtool.
Also had to run cleaning tape thru DLT drive 2X to clear the "clean me"
request during the reboot.
___end___
COMMENT: Tue Jul 12 00:48:12 GMT 2005
Shutting down in heavy clouds.  Going back to GONG on my way down to
pack up the flaky hard drive and the UPS they sent out earlier.  
Will try to get down to FEDEX before they close for the day.
___end___
Tue Jul 12 00:50:23 GMT 2005 
        MkIV    
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